Calculus

MA 141 Fall 2000

Instructor:

ffl Amy Langville ffl 513-1907 ffl 104 Daniels (in the basement) ffl anlangvi@eos.ncsu.edu ffl Office Hours: TW 9:30-11am

Class: meets MTWF in HA 335 Lab: meets in HA 314 for 5 Thursdays during the semester (9/14, 9/21, 10/5, 11/2, 11/9). Text: CALCULUS Concepts and Contexts by James Stewart, Brooks/Cole, 1999, NC State Univer-

sity Edition, Price: $120

Homework: There will be regular class homeworks due every other week. There will also be 5 Maple

homeworks to be submitted electronically. Maple questions, help and information can be found at:

www2.ncsu.edu/eos/info/maple_info/www/maple_site_summer_2000/index.html

Writing Assignments: There will be three optional writing assignments. Each writing assignment, if

completed successfully, will count as 5% of your total grade. If all 3 assignments are completed, then this 15% of your grade can replace part of your final exam grade. Thus your final exam grade will only be 15% of your overall grade in the class, rather than 30%. (See Writing Assignment directions and samples.)

Attendance: Four or more absences and your grade will be docked accordingly. Grading: The University plus/minus system will be used. The breakdown will be as follows:

ffl Exams 50% ffl Homework (Class and Maple) 15% ffl Final Exam 30% ffl Attendance and Participation 5%

Schedule: The day-to-day schedule for MA 141 should be available soon at the above site by clicking

on the Course Syllabi link.